Author’s response to reviews

Title: Abscess of adrenal gland caused by disseminated subacute Nocardia farcinata pneumonia. A case report and mini-review of the literature

Authors:

Michael Tachezy (mtachezy@uke.uni-hamburg.de)
Philipp Simon (psimon@uke.uni-hamburg.de)
Corina Ilchmann (c.ilchmann@uke.de)
Yogesh K Vashist (vashist@uke.uni-hamburg.de)
Jacob R Izbicki (izbicki@uke.de)
Karim A Gawad (gawad.karim@hohg.de)

Version: 4 Date: 5 November 2009

Author’s response to reviews: see over
Major revisions
Abstract - Please divide this into the following sections: Background, Case Presentation, Conclusions
Structure: this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Case Presentation; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Authors' information (optional); Acknowledgements (optional); References
Authors' contributions: Please place an Authors' contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.
Please use the following format as an example, including using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

Minor revisions
Line numbers - Please remove these.
Keywords - Please remove these.
Acknowledgements: If you have any acknowledgements, please place an Acknowledgements section at the end of the manuscript text, directly preceding the Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.
Figure citations - Please be consistent when referring to figures, using either 'Figure' or 'Fig.' throughout the manuscript and figure legends.
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.